Author's response to reviews

Title: Individual, Social Environmental, and Physical Environmental Correlates with Physical Activity among Canadians: a Cross-sectional Study

Authors:

Sai Yi Pan (sai_yi_pan@phac-aspc.gc.ca)
Christine Cameron (ccameron@cflri.ca)
Marie DesMeules (marie_desmeules@phac-aspc.gc.ca)
Howard Morrison (Howard_Morrison@phac-aspc.gc.ca)
Cora Lynn Craig (ccraig@cflri.ca)
Xiaohong Jiang (Xiaohong_Jiang@phac-aspc.gc.ca)

Version: 3 Date: 22 October 2008

Author's response to reviews: see over
October 21, 2008

RE: MS: 2056753089207845 - Influences of Individual, Social Environmental, and Physical Environmental Correlates with Physical Activity Participation among Canadians: A Cross-sectional Study

Dear Editors,

Thank you very much for your e-mail of October 7, 2008 inviting us to submit a further revised version of above-mentioned manuscript. We have revised the manuscript according to reviewer’s comments, including editing and improving the style of the written English and being consistent in the use of some terminologies. We have also checked the manuscript to make sure that it conforms to the journal’s format and included a “Acknowledgements” section. We hope that the revised manuscript is now suitable to be published in BMC Public Health.

In addition, we would like a press release about this article.

Sincerely yours,

Sai Yi Pan, M.D.
Public Health Agency of Canada